Author's response to reviews

Title: Did HealthKick primary school nutrition intervention improve dietary diversity and decrease intake of unhealthy snacks by children in low-income settings in South Africa?

Authors:

Nelia P Steyn (nelia.steyn@uct.ac.za)
Anniza de Villiers (anniza.devilliers@mrc.ac.za)
Nomonde E gwebushe (nomonde.gwebushe@mrc.ac.za)
Catherine Draper (catherine.draper@uct.ac.za)
Jillian Hill (jillian.hill@mrc.ac.za)
Mariena de Waal (mdewaal@fedicsunit.co.za)
Lucinda Dallais (luc1_za@yahoo.com)
Zulfa Abrahams (zabrahams@brhc.com)
Carl Lombard (carl.lombard@mrc.ac.za)
Estelle V Lambert (vicki.lambert@uct.ac.za)

Version: 2 Date: 26 February 2015

Author's response to reviews:

1. All authors details were provided as requested.
2. Competing interests section was included
3. Detailed information on authors contributions were provided